Author’s response to reviews

Title: Squamous cell carcinoma (Marjolin’s ulcer) in an oro-cutaneous fistula of a large mandibular ameloblastoma: a case report.

Authors:

Peter M Nthumba (nthumba@gmail.com)

Version: 6 Date: 26 May 2011

Author’s response to reviews: see over
Prof Michael Kidd,

Editor-in-Chief,

Journal of Medical Case Reports.

Dear Sir,

I hereby resubmit this manuscript, ‘Squamous cell carcinoma (Marjolin’s ulcer) in an oro-cutaneous fistula of a large mandibular ameloblastoma: a case report.’ (MS: 1595699943539710) for consideration and possible publication in the Journal of Medical Case Reports.

As requested, Figures 1 and 5 have been cropped down to the nose, without loss of pathology detail.

Pleasantly, the patient showed up at thirteen months post-operatively, and Figure 5 and the text have been updated to reflect the same.

Thank you.

Peter M. Nthumba, MD

AIC Kijabe Hospital

26th May 2011